Author's response to reviews

Title: A Patient with Acute Abdomen due to Metastatic Small Bowel Melanoma: a case report

Authors:

Georgios Lianos Mr (georgiolianos@yahoo.gr)
Rodamanthos Doumos Mr (broutos@yahoo.it)
Thomas Messinis Mr (tomessinis@yahoo.gr)
Christina Bali Mrs (cbali03@yahoo.com)

Version: 2 Date: 3 May 2013

Author's response to reviews: see over
Cover Letter by G. Lianos

Paper revised in accordance with the recommendations of the editorial team of the journal and with the reviewer’s changes. All revised points are highlighted by green colour in the manuscript.

-I have changed the title

-I have changed the first page

-I have the abstract in page 2 with the structure introduction, case presentation, conclusion. change in line 2.

-Change key words

-Introduction: change in line 1,2,5,6,7

-Case report: change in line 1,4,6, I have reported the lab.values, I have added the number of the figure and I have reported the histologic report

-Discussion: change in line 2,4, completely re-structure the last lines as recommended

-Conclusion: change line 1,2,3

- References in accordance with the instructions

- Changes in Authors’ Contribution